Author's response to reviews

Title: Evidence-based patient choice: a prostate cancer decision aid in plain language

Authors:

Margaret Holmes-Rovner (mholmes@msu.edu)
Sue Stableford (sstableford@une.edu)
Angela Fagerlin (fagerlin@med.umich.edu)
John T. Wei (jtwei@umich.edu)
Rodney L. Dunn (rdunn@umich.edu)
Janet Ohene-Frempong (frempongjl@aol.com)
Karen Kelly-Blake (kellyka1@msu.edu)
David R. Rovner (rovner@msu.edu)

Version: 8 Date: 9 June 2005

Author’s response to reviews:

Elizabeth C. Moylan, PhD
Assistant Editor
BioMed Central

Dear Dr. Moylan:

Thank you for your recent e-mail and especially for the cogent specific recommendations. I have also read the comments of Dr. Kirsten McCaffery.

The changes made include:

1. Revised table 3 partially. It is impossible to move the number of replies to the heading in the historical controls since there are 3 different numbers of replies.
2. Changed all p values to 2 decimal places.
3. Changed author list.
4. Changed support details.
5. Removed keyword and word count sections.
6. Renamed sections in accord with template provided.
7. Changed method of citing multiple references and moved before full stop.
8. Moved website references out of text.
9. Changed method of referring to web sites.
11. Deleted the additional file.

I hope the manuscript is now acceptable for publication. We are delighted to see this appear in "print" in your journal. We feel it is a ground breaking article in the field.

Sincerely yours,
Margaret Holmes-Rovner, PhD